

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Gerlagh, Reyer; Jaimes, Richard; Motavasseli, Ali

## Working Paper Global Demographic Change and Climate Policies

CESifo Working Paper, No. 6617

**Provided in Cooperation with:** Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

*Suggested Citation:* Gerlagh, Reyer; Jaimes, Richard; Motavasseli, Ali (2017) : Global Demographic Change and Climate Policies, CESifo Working Paper, No. 6617, Center for Economic Studies and ifo Institute (CESifo), Munich

This Version is available at: https://hdl.handle.net/10419/171081

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



## WWW.ECONSTOR.EU



# Global Demographic Change and Climate Policies

Reyer Gerlagh, Richard Jaimes, Ali Motavasseli



#### Impressum:

CESifo Working Papers ISSN 2364-1428 (electronic version) Publisher and distributor: Munich Society for the Promotion of Economic Research - CESifo GmbH The international platform of Ludwigs-Maximilians University's Center for Economic Studies and the ifo Institute Poschingerstr. 5, 81679 Munich, Germany Telephone +49 (0)89 2180-2740, Telefax +49 (0)89 2180-17845, email <u>office@cesifo.de</u> Editors: Clemens Fuest, Oliver Falck, Jasmin Gröschl www.cesifo-group.org/wp

An electronic version of the paper may be downloaded

- · from the SSRN website: <u>www.SSRN.com</u>
- from the RePEc website: <u>www.RePEc.org</u>
- from the CESifo website: <u>www.CESifo-group.org/wp</u>

## **Global Demographic Change and Climate Policies**

## Abstract

Between 1950 and 2017, world average life expectancy increased from below-50 to above-70, while the fertility rate dropped from 5 to about 2.5. We develop and calibrate an analytic climate-economy model with overlapping generations to study the effect of such demographic change on capital markets and optimal climate policies. Our model replicates findings from the OLG-demography literature, such as a rise in households' savings, and a declining rate of return to capital. We also find that demographic change raises the social cost of carbon, at 2020, from 28 euro/tCO<sub>2</sub> in a model that abstracts from demography, to 94 euro/tCO<sub>2</sub> in our calibrated model.

JEL-Codes: H230, J110, Q540, Q580.

Keywords: climate change, social cost of carbon, environmental policy, demographic trends.

Reyer Gerlagh Department of Economics Tilburg University / The Netherlands R.Gerlagh@uvt.nl

Richard Jaimes Department of Economics Tilburg University / The Netherlands R.V.JaimesBonilla@uvt.nl Ali Motavasseli Department of Economics Institute for Management and Planning Studies / Tehran / Iran a.motavasseli@imps.ac.ir

August 16, 2017 Link to most recent version:

www.dropbox.com/s/wmfmb0uljgunokq/demographics\_climate\_policies\_webversion.pdf?dl=0 We are grateful for helpful comments and suggestions from Inge van den Bijgaart, Sjak Smulders, Matti Liski, participants at the CESifo Climate and Energy workshop, the University of Gothenburg Economics Department seminar, and the EAERE-FEEM-VIU European Summer School. All errors are ours.

## 1. Introduction

Between 1950 and 2017, world average life-expectancy increased from below-50 to above-70, while the fertility rate dropped from 5 to about 2.5 children per woman (Figure 1). The worldwide rise of life-expectancy and fall of total fertility rate is expected to robustly continue for the remainder of the current century, with the emerging economies catching up the patterns typical for economies that industrialized before. The worldwide demographic trends decreases the ratio of young relative to old and change the propensity to save of the average consumer. Both supply of labour and capital will adjust, and future capital returns will most likely differ from those in the past.



Note: The year 2017 is indicated with a vertical line. Source: United Nations. World Population Prospects. Medium-variant projections. The projections do not take into account projected climate change, but the expected effect is small compared to other major drivers of demographic change.

Climate policies search the optimal trade-off between consumption, investments in man-made capital, and investment in natural capital through emissions reductions (Nordhaus, 1993). How should climate policies adjust in response to global demographic change? Using a stylized overlapping generations model (based on Fanti and Gori (2012); Cipriani (2014)), we analytically and quantitatively analyze the effects of an increase in life-expectancy and a decrease in fertility rates on households' savings, aggregate capital investments, and efficient climate policies. For comparison we also study demography and optimal climate policy in a Ramsey-Cass-Koopmans model (as in Nordhaus (1993)). The overlapping generations (OLG) model is the standard approach to study demography's effects on savings and investments. The Ramsey-Cass-Koopmans (RCK) model, with an infinitely-lived representative agent, is the most common type of model to study climate policy in the context of economic growth. We compare the effects of demography on climate policy in both types of models. The OLG model suggests that falling fertility and increasing life expectancy leads to a robust increase in future households savings. In turn, these lead to a substantial tightening of *present* efficient climate policies, that is, a large increase in the (present) efficient carbon price.<sup>1</sup> The reason that future capital markets have such robust effects for present climate policies comes from the extreme persistence of climate change (Gerlagh and Liski, 2017b). The same demographic patterns, in the RCK model, result in a decrease in both the future savings rate and the present carbon price. We show that the OLG and RCK types of model thus provide radically different outcomes. Therefore, we call for a closer look at previous quantitative estimates for the social costs of carbon (SCC), which mostly neglected the effects of demographic patterns on capital markets and climate policies.

The intuition for our main results, derived with the OLG model, is straightforward. The two worldwide demographic trends, increasing life expectancy and decreasing fertility, both increase household savings. First, since people expect to live longer, they require more resources when old after they leave the labour market.<sup>2</sup> Second, a decrease in the fertility rate causes an increase in labor supply when young, and thus, a rise in income, which is partly saved to smooth consumption over the life-cycle. As capital stocks increase, the returns fall, and it becomes relatively more attractive to invest in climate change mitigation.

Our analytical model presents an analytic formula for the social cost of carbon, connecting to a large literature. A large literature on the social costs of carbon is devoted to the question of the discount rate by which future climate damages are discounted before aggregation in the Net Present Value (Weitzman, 2013).<sup>3</sup> Both the so-called 'pure' discount rate for utility streams and the discount rate for consumption goods feature

<sup>&</sup>lt;sup>1</sup>In this paper, we use the terms (efficient) carbon price and the social costs of carbon (SCC) interchangeably. They refer to the net present value of damages associated with one extra unit of CO2 emissions.

<sup>&</sup>lt;sup>2</sup>Our OLG model base set up is a fully funded pension system. We do not examine the effects of changes in institutions that affect households' transfer of resources from their young age to their old age, e.g., the pension system.

<sup>&</sup>lt;sup>3</sup>See Guo et al. (2006) for an analysis of the consequences of declining discount rates on estimates of the social cost of carbon. See also Karp and Traeger (2013) and Weitzman (2013) for a discussion about discounting in cost-benefit analysis and Mastrandrea and Schneider (2001) for a better understanding of the effects of relatively low discount rates on climate policies. van den Bergh and Botzen (2015) present a recent survey of this literature.

prominently in this literature. Recently, van den Bijgaart et al. (2016) presented an alternative discount rate measure. They show that the relative value of future output compared to present output,  $b_t = \frac{Y_{t+1}}{r_{t+1}Y_t}$  (labeled the effective discount factor), where  $Y_t$  is output and  $r_{t+1}$  is the marginal rate of substitution between the two periods, captures the most important discount information required to calculate the efficient carbon price.<sup>4</sup> We use the effective discount factor to gauge the implications of our approach for the SCC.

Table 1 reports the values of the key parameters and the implied effective annual discount factors  $b_t$  used in previous studies. Before reading the table, we note that in most models, the effective discount factor varies with demography. To the best of our knowledge, this is the first paper that makes this connection explicit. The previous studies reported in Table 1 rely on the Ramsey-Cass-Koopmans (RCK) model, in which the Ramsey rule,  $r_t = \beta^{-1} m_t^{1-\xi} \gamma_t^{\eta}$ , connects the marginal rate of substitution,  $r_t$ , to the pure time preference factor  $\beta$ ,<sup>5</sup> population growth  $m_t$ , where  $\xi \in \{0, 1\}$  is the population weight in the social planner problem,<sup>6</sup> the elasticity of marginal utility  $\eta$  and the per capita growth factor of consumption  $\gamma_t$ . The effective discount factor  $b_t$  then also depends on the scale of the economy through both population growth  $m_t$  and productivity growth, which approximately also measures per capita consumption growth,  $Y_{t+1}/Y_t \approx m_t \gamma_t$ , so that  $b_t \approx m_t \gamma_t / r_t = \beta m_t^{\xi} \gamma_t^{1-\eta}$ .

As noted in the literature, the effective discount factor is sensitive to parameter setting procedures. For instance, whereas Stern (2007), following an ethics-based approach, chooses a rate of pure time preference of 0.1% per year, Nordhaus (2008), Nordhaus (2014) and Golosov et al. (2014), using a market-based approach, set a rate of 1.5%. Higher values for this parameter can be found in Weitzman (2007) and Nordhaus (1993) (first column entries of Table 1). Importantly, comparing the columns  $b_t$  for 2010 and 2100, we see that because population growth,  $m_t$ , is expected to decrease in the coming century, in the standard RCK model, the implied effective discount factor,  $b_t$ , does not

<sup>&</sup>lt;sup>4</sup>See eq (18) in this paper.

<sup>&</sup>lt;sup>5</sup>We can switch between rates and factors, through  $\beta = 1/(1 + \rho)$  where  $\rho$  is the pure time preference rate.

<sup>&</sup>lt;sup>6</sup>For Benthamite welfare weighing utility by population size,  $\xi = 1$  and the rate of substitution only depends on per capita consumption growth. For Millian welfare aggregating average utility,  $\xi = 0$  and the rate of substitution increases with population growth. See Canton and Meijdam (1997) for more details.

<sup>&</sup>lt;sup>7</sup>We use an approximate sign because equality requires that the consumption share is constant, which is approximately correct.

depend on population growth directly as in the RCK model, but only indirectly through its effects on life-cycle savings. The effective discount factor varies with demographic factors, most importantly life-expectancy, through changes in demand for capital as a retirement investment. To relate our results to the existing literature, we calibrate the model parameters such that the (time-varying) effective annual discount factor at 2010 equals the constant discount factor b = 0.985 used in Golosov et al. (2014). As the table shows, we find the effective discount factor to increase over the next century, rather than to decrease.

	β	$\eta$	ξ	$b_t$		$SCC_t$	
				2010	2100	2010	2100
Nordhaus (1993)	0.970	1	1	0.982	0.971	14.2	110
Weitzman (2007)	0.980	2	1	0.974	0.962	10.7	84.1
Stern (2007)	0.999	1	1	1.011	1.000	-	-
Nordhaus (2008)	0.985	2	1	0.979	0.967	12.5	96.3
Nordhaus (2014)	0.985	1.45	1	0.989	0.978	18.7	140
Golosov et al. (2014)	0.985	1	0	0.985	0.985	21.2	206
This paper	n.a.	1	n.a	0.985	0.996	63.8	794

Table 1: Effective annual discount factors and social costs of carbon

Note:  $\beta$ : the discount factor for utility between generations. Its value is exogenous in Ramsey-Cass-Koopmans models, and endogenous in our model.  $\eta$ : the elasticity of marginal utility (the inverse of the elasticity of intertemporal substitution).  $\xi$ : indicator parameter for weighting utility.  $b_t \approx \beta m_t^{\xi} \gamma_t^{1-\eta}$ : the effective discount factor. For per capita consumption growth we use  $\gamma = 1.019$  for 2010 and  $\gamma = 1.02$  for 2100. For population growth, we use m = 1.012 for 2010 and m = 1.001 for 2100.  $SCC_t$ : The social costs of carbon [EUR/tCO<sub>2</sub>] tends to increase with  $b_t$  and with income, see eq (18), and Appendix C for the quantitative calculation of  $b_t$ .

The quantitative effects on the social costs of carbon, of these divergent patterns for the discount factor, are shown in the last two columns of Table 1. For instance, at our calibration year 2010, a model that features no demographics as described in Golosov et al. (2014) yields a carbon price 67% lower than the one related to our benchmark model. This difference is even higher when our OLG model is compared with the other Ramsey-Cass-Koopmans models that impose a negative effect of decreasing population growth rates on optimal savings.

## 2. Further literature

Our paper relates to three strands of literature.<sup>8</sup> Firstly, there are a few numerical Integrated Assessment Models (IAMs) that study climate policies in an economy with overlapping generations (OLG). Howarth (1998) and Gerlagh and van der Zwaan (2001) are the closest to this paper. Recently, Quaas and Bröcker (2016) developed an analytical OLG-climate model.<sup>9</sup> We extend these papers by providing a rich set of analytic results that characterize formally the effect of demography on climate policies. We add a formal analysis of demography in the (competing) Ramsey-Cass-Koopmans (RCK) model type. Furthermore, we set up a calibration procedure that quantifies the effects through a transparent approach, as compared to the quantitative IAMs with many moving parts in which causality is hard to gauge.

Secondly, we relate to recent economic research suggesting that demographic patterns over the last decades have increased aggregate household savings rates and decreased the capital returns, mainly through shifts in the age structure of the population (Canton and Meijdam, 1997; Bloom et al., 2003, 2007; Eggertsson and Mehrotra, 2014; Curtis et al., 2015; Carvalho et al., 2016; Imrohoroglu and Zhao, 2017). Demography also drives part of international trade and capital flows, which reflect discrepancies between savings and investments across countries (Backus et al., 2014; Fedotenkov et al., 2014; Krueger and Ludwig, 2007). Demographic change has been connected to housing prices, since lower returns to capital in the future create incentives for seeking alternative stores of value (Chen and Wen, 2017), and changes in social security systems due to population aging (Imrohoroglu and Zhao, 2017; Cipriani, 2014; Fanti and Gori, 2012). Curtis et al. (2015) show that demographic patterns are able to explain over half of the household savings rate fluctuations in China, while Sánchez-Romero (2013) finds that the demographic transition in Taiwan, during the period 1965-2005,

<sup>&</sup>lt;sup>8</sup>Notice that in our modelling, demographic variables are exogenous and temperature changes only affect production. Several studies assess the effect of population aging on emissions and the environment, see Balestra and Dottori (2012), Jouvet et al. (2010), Varvarigos (2010), Mariani et al. (2010), and Ono (2005). De la Croix and Gosseries (2012) study the effect of climate policies on fertility and human capital. In a similar vein, with respect to the relationship between carbon emissions and population, Casey and Galor (2017) provide empirical evidence that a reduction in fertility rates, while increases income per capita, diminishes carbon emissions. Likewise, Dalton et al. (2008) provides an analysis of the effect of population aging on U.S energy use and carbon emissions. None of these studies point to the impact of demographic patterns on savings rates and, subsequently, on climate policies.

<sup>&</sup>lt;sup>9</sup>Karp and Rezai (2014) use an OLG model for a somewhat different purpose. They show that if the environmental benefits of climate policies can be capitalized in an increased value of the capital stock, current generations also benefit from environmental policies.

substantially increased the investment rate, leading to more capital and higher GDP per capita growth.<sup>10</sup> Going forward, while our model formulation can reproduce the results mentioned above, we add to this literature the impact of changing savings, due to population dynamics, on the evaluation of expected future environmental damages and climate policies.

Thirdly, our theoretical framework also builds on an emerging analytical literature about optimal carbon taxes as dependent on a few parameters: pure time preferences, a carbon cycle structure, temperature adjustments, and expected damages due to climate change (Golosov et al., 2014).<sup>11</sup> Others have added risk updating when climate information arrives (Gerlagh and Liski, 2017a), temperature delays and general time preferences (Gerlagh and Liski, 2017b),<sup>12</sup> and tax distortions (Barrage, 2016). We add demographic parameters to these.

The remainder of this paper proceeds as follows. Section 3 presents the benchmark model and describes the equilibrium concept. Section 4 discusses existence issues and provides a comparative statics analysis. Section 5 introduces the RCK model with demography and outlines optimal climate policies. Section 6 includes a quantitative assessment and discusses parameter calibration, data, and the climate module, to calculate the social cost of carbon. Section 7 concludes.

## 3. The model

We consider a two-period OLG model with full depreciation and log utility based on Cipriani (2014) and Fanti and Gori (2012) to analyze the impact of changes in fertility and life-expectancy on savings and, therefore, through the aggregate capital market, the effect of the demographic transition on environmental policies. Time is discrete and runs to infinity. The economy's technology is based on Gerlagh and Liski (2017b), while our quantitative assessment employs the (median) climate-damage description listed in van den Bijgaart et al. (2016).

<sup>&</sup>lt;sup>10</sup>It is important to mention that we model a closed economy and therefore the aggregate savings rate equals the investment rate.

<sup>&</sup>lt;sup>11</sup>For a discussion of the implications and robustness of simple carbon pricing rules see e.g., Barrage (2014), van den Bijgaart et al. (2016), and Rezai and van der Ploeg (2015).

<sup>&</sup>lt;sup>12</sup>See also Iverson (2012); Belfiori (2013) for discussions on time preferences and the SCC formula.

#### 3.1. Households

Each agent lives with certainty during the first period, and faces a probability  $h_t$  of surviving to the second period. Individuals work when young and consume their savings when old. Consumption by children is not explicitly modeled, but the time that parents spend raising their children,  $\chi n_t < 1$ , where  $n_t$  is the number of children and  $\chi$  is the time per child, is deducted from labour supply. That is, the size of each generation is denoted by  $N_t$ , with  $N_{t+1} = n_t N_t$ , and  $M_t = N_t + h_{t-1} N_{t-1}$  is the total population. Only the young generation works, and the labour supply is inelastic, given by

$$L_t = (1 - \chi n_t) N_t. \tag{1}$$

The household chooses consumption and savings to maximize life-time utility,

$$\max_{\{c_{1,t},c_{2,t+1},s_t\}} u_t = u_{1,t} + \delta h_t u_{2,t+1},\tag{2}$$

where  $u_{1,t} = \ln(c_{1,t})$ ,  $u_{2,t} = \ln(c_{2,t+1})$ , subject to the budget constraints when young and old,

$$c_{1,t} + s_t + \chi w_t n_t = w_t + \tau_t,$$
 (3)

$$h_t c_{2,t+1} = r_{t+1} s_t, (4)$$

where  $c_{1,t}$  and  $c_{2,t+1}$  denote consumption at young and old age (when living);  $s_t$  are savings,  $\delta \in (0, 1)$  refers to the discount factor;  $h_t \in (0, 1)$  is the probability of surviving (longevity) which varies exogenously over time, so that  $h_t c_{2,t+1}$  is expected consumption when old. The variable  $r_{t+1}$  is the return to savings;  $w_t$  represents the wage that individual receives for supplying inelastically one unit of labour;  $\tau_t$  is the per capita government lump-sum transfer (recycling of carbon tax revenues, discussed below),  $n_t$ is the exogenous fertility rate and  $\chi \in (0, 1)$  denotes the fraction of time that it is needed for raising children. Note that we study the effects of (exogenous) demography on climate policies. By considering endogenous fertility and health choices, as in Becker and Lewis (1973), the analysis can be extended to describe demographic policies as part of climate change policies (Harford, 1998; Murtaugh and Schlax, 2009; Bohn and Stuart, 2015).<sup>13</sup>

<sup>&</sup>lt;sup>13</sup>These papers do not look at the effect of demography on climate policies through capital markets.

The first-order conditions give the usual Euler equation:

$$\frac{c_{2,t+1}}{c_{1,t}} = \delta r_{t+1}.$$
(5)

Using the Euler equation and the budget constraints (3,4), the savings of households can be derived as

$$s_t = \frac{\delta h_t}{1 + \delta h_t} (w_t (1 - \chi n_t) + \tau_t).$$
(6)

For analysis, we assume that demography converges to some long-term trend.

**Assumption 1.** Fertility and life-expectancy converge to long-term values:  $n_t \rightarrow n_{\infty} > 0$ ,  $h_t \rightarrow h_{\infty} \leq 1$ .

#### 3.2. Production and Climate

Following Gerlagh and Liski (2017b), we model the economy's technology, which is a function of capital, labour, fossil fuel energy, and the past emissions, in the following way:

$$Y_t = F(K_t, E_t, L_t; z_t), \tag{7}$$

with constant returns to scale in capital  $K_t$ , labour  $L_t$ , and emissions  $E_t$  (proximated by use of fossil fuel energy sources), and  $z_t = (E_{t-1}, E_{t-2}, ...)$  is the history of emissions. Production has constant elasticity in capital, and climate change damages are described through a multiplicative factor,

$$F(K_t, E_t, L_t; z_t) = \Omega(z_t) K_t^{\alpha} [A_t(E_t, L_t)]^{1-\alpha},$$
(8)

with  $\alpha \in (0, 1)$ , and  $\Omega(z_t)$  refers to output loss due to past emissions.  $A_t(E_t, L_t)$  has constant returns to scale, and is time dependent, accounting for effective energy-labour combination. When convenient, we write  $a_t \equiv A_t/L_t$  as labour-augmenting productivity, which only depends on per labour emissions,  $a_t = a_t(e_t)$ , with  $e_t \equiv E_t/L_t$ . One can rearrange the production function into per effective labour terms to get

$$y_t = \Omega(z_t) k_t^{\alpha} a_t^{1-\alpha}, \tag{9}$$

where  $y_t \equiv Y_t/L_t$  and  $k_t \equiv K_t/L_t$  are output- and capital-labour ratios, respectively.

**Assumption 2.** Effective labour productivity  $a_t(e_t)$  is continously differentiable, increasing in emissions per labourer, strictly concave, emissions are not essential and become zero for a finite carbon price,  $a'_t > 0$  and  $a''_t < 0$  for  $e_t < e_t^{\max}$ , for some  $e_t^{\max} > 0$ ;  $a'_t(e_t^{\max}) = 0$ ,  $a_t(0) > 0$ ,  $a'_t(0) < \infty$ .

We keep a general climate impact structure. Golosov et al. (2014) relate climate impacts to atmospheric  $CO_2$  stocks, while Gerlagh and Liski (2017a) provide an in-depth discussion of the impact function based on delays in temperature change. The general formulation is

$$\Omega(z_t) = \exp(-\sum_{i=1}^{t-1} \theta_i E_{t-i}), \qquad (10)$$

where  $\theta_i$  can be interpreted as a representation of the carbon cycle and temperature adjustments (see van den Bijgaart et al. (2016)).

Each period, a representative firm uses capital K, labour L and energy E to produce a final good using a Cobb-Douglas technology and solves the following problem:

$$\max_{\{K_t, L_t, E_t\}} F(K_t, E_t, L_t; z_t) - r_t K_t - w_t L_t - p_t E_t,$$
(11)

where  $p_t$  refers to the carbon emissions price, and the firm considers  $\Omega_t$  as exogenous. The first-order conditions imply

$$r_t = \alpha \frac{Y_t}{K_t},\tag{12}$$

$$w_t = (1 - \alpha) Y_t \frac{A_{L,t}}{A_t},\tag{13}$$

$$p_t = (1 - \alpha) Y_t \frac{A_{E,t}}{A_t}.$$
(14)

where  $A_{X,t}$  is the derivative of  $A_t$  with respect to  $X_t$ . We have that labour income plus government transfers are a constant share of output,

$$w_t L_t + p_t E_t = (1 - \alpha) Y_t.$$

#### 3.3. Carbon policies

The carbon regulator can set carbon taxes  $p_t$ , and lump-sum income transfers  $\tau_t$ . The regulator is subject to an intertemporal budget condition: the net present value of car-

bon tax revenues equals lump-sum transfers,

$$\sum_{t=1}^{\infty} \frac{N_t \tau_t}{r_1^t} = \sum_{t=1}^{\infty} \frac{p_t E_t}{r_1^t},$$
(15)

where  $r_t^{t+i}$  is the capital return between period t and t + i,  $r_t^{t+i+1} \equiv r_t^{t+i}r_{t+i+1}$ . As a central case, detailed in the next section, we consider equilibria where the government budget balances within each period,

$$N_t \tau_t = p_t E_t. \tag{16}$$

In Section 4.1 we derive equilibrium rules for carbon pricing  $p_t$ . Here, we define the market costs of carbon as the net present value of future marginal damages evaluated at market prices,

$$MCC_t = \sum_{i=1}^{\infty} \frac{1}{r_t^{t+i}} \frac{\partial Y_{t+i}}{\partial \Omega_{t+i}} \frac{-\partial \Omega_{t+i}}{\partial E_t}.$$
(17)

When we substitute (10) in (8), we find

$$MCC_t = \sum_{i=1}^{\infty} \frac{\theta_i Y_{t+i}}{r_t^{t+i}} = Y_t \sum_{i=1}^{\infty} \theta_i b_t^{t+i},$$
(18)

where  $b_t^{t+i}$  is the net present value of future output, at t + i, relative to current output, defined recursively through  $b_t^{t+1} \equiv b_t$ ,  $b_t^{t+i} \equiv b_t b_{t+1}^{t+i}$ , and

$$b_t \equiv \frac{Y_{t+1}}{r_{t+1}} \frac{1}{Y_t}.$$
(19)

Formula (18) for the carbon price is similar to carbon price formulas presented in Golosov et al. (2014), van den Bijgaart et al. (2016), and Gerlagh and Liski (2017b). The carbon price tends to increase proportionally with output as a larger economy values emissions more. In a larger economy, there is more at stake as damages also scale with the economy. Thus, an increase in labour supply, complemented with a proportional increase in capital, will tend to raise output, and the carbon price alongside. If we label the summation term on the RHS of (18) as  $g_t$ , then it reads  $MCC_t = g_tY_t$ . In the expression,  $g_t$  depends on the value of future output, but we will provide conditions when  $g_t$  becomes a constant, that is, independent of the current capital stock or future policies. Conditional on Assumption 2, and given capital and labour factor supply, the next lemma establishes a unique static equilibrium in case  $g_tY_t$  is implemented as the

carbon price.

**Lemma 1.** For any given  $L_t > 0$ , and  $0 \le g_t \le (1 - \alpha) \frac{A_{E,t}(0,L_t)}{A_t(0,L_t)}$ , there is a unique  $E_t \ge 0$  such that

$$(1-\alpha)\frac{A_{E,t}(E_t, L_t)}{A_t(E_t, L_t)} = g_t$$

In this economy, emissions versus output tend to follow a so-called Environmental Kuznets curve. For low output levels (low levels of factor supply), carbon prices are negligible and emissions increase approximately proportional with factor supply and output. For high output (factor supply), the term  $\frac{A_{E,t}(0,L_t)}{A_t(0,L_t)}$  falls below  $g_t$ , and emissions become zero.

Below, we will establish closed-form solutions for  $b_t$ , and conditions under which optimal carbon prices equal the market valuation of carbon,  $p_t^* = MCC_t$ .<sup>14</sup> Novel in our analysis is that we study how the discount factor  $b_t^{t+i}$  varies with time as a result of demographic change. Therefore, optimal climate policies adjust to demographic changes as well; (18) becomes demography-dependent. We then compare our solution for the carbon price with a closed-form solution for a Ramsey-Cass-Koopmans model with Benthamite welfare as in Nordhaus (2008) and most other Integrated Assessment Models, and with Millian welfare as in Golosov et al. (2014).

#### 3.4. Equilibrium

In equilibrium, aggregate consumption plus investments equal production,

$$N_t c_{1,t} + h_{t-1} N_{t-1} c_{2,t} + K_{t+1} = Y_t.$$
(20)

We can now define the competitive equilibrium as follows:

**Definition 1.** Given the parameter set  $\{\chi, \delta, \alpha, (\theta_i)\}$ , a competitive equilibrium is an allocation  $\{c_{1,t}, c_{2,t}, s_t, Y_t, K_{t+1}, E_t, \tau_t\}_{t=1}^{\infty}$ , supported by prices  $\{r_t, w_t, p_t\}_{t=1}^{\infty}$  such that (i) households maximize their life-time utility, (3-5), (ii) firms maximize profits, (12-14), (iii) the government satisfies the inter-temporal balanced budget, (15), and (iv) the aggregate goods market clearing condition holds, (20).

The definition leaves open how carbon prices are determined; it does not impose carbon prices to reflect the social costs of carbon. Notice that the definition also does not

<sup>&</sup>lt;sup>14</sup>We do not consider tax-interaction effects as in Barrage (2016).

consider conditions about income transfers between generations. The first refinement of the equilibrium concept specifies that the regulator does not transfer income between generations. We define:

**Definition 2.** *A competitive equilibrium is called 'representative' if the government satisfies its budget constraint period-by-period, i.e.* (16) *holds.* 

The definition captures the idea that we want to study carbon taxes in isolation; we want carbon policies to be free from a potential preference for redistributing wealth across generations. The representative equilibrium defines what it means to exclude redistributional preferences.<sup>15</sup> Condition (16) determines the welfare weights, defined shortly below. They depend on demography, but are the same in a representative equilibrium with or without climate damages, and with or without climate policies.

This 'representative' equilibrium is the OLG-equivalent of the infinitely-lived representative agent approach in RCK models such as DICE (Nordhaus, 2008). The typical RCK model chooses its social preference parameters such that, in absence of a climate externality, the model reproduces realistic consumption smoothing and capital investment rates. Similarly, in our model, we calibrate preferences  $\delta$  such that the competitive equilibrium without intergenerational income transfers generates realistic macroeconomic savings rates.<sup>16,17</sup>

There is a connection between inter-generational income transfers  $\tau_t$ , capital investments  $K_{t+1}$  and the value ratio  $b_{t+1}$ . We can rewrite the inter-temporal government budget (15) through a public savings variable  $S_t^g$  as

$$S_t^g = r_t S_{t-1}^g + p_t E_t - N_t \tau_t,$$
(21)

<sup>&</sup>lt;sup>15</sup>Compare with an economy that has no climate externality, i.e.  $\theta = (0, 0, ...)$ . The *representative laissezfaire* competitive equilibrium,  $\tau_t = p_t = 0$ , is then efficient and does not redistribute income.

<sup>&</sup>lt;sup>16</sup>We could say that the regulator in our economy derives its implicit welfare weights,  $\beta_t$  defined below, from historic macro-economic savings. As we see below in equations (22) and (23), our model also has an equivalent to 'low discounting' in the RCK model. So-called 'low' pure discount rates in RCK models are often considered fair from an ethical perspective, but discarded as inconsistent with revealed social preferences, because they would result in unrealistically high investment rates. In our economy, the equivalent of 'low discounting' is an equilibrium with positive income transfers to future generations,  $S_t^g > 0$ . Such transfers would result in artificial 'high' investment rates at the macro-economy level and higher implicit welfare weights for future generations.

<sup>&</sup>lt;sup>17</sup>Technically, the representative equilibrium is similar to the use of Negishi welfare weights in a multicountry model. As an example, the multi-region RICE model (Nordhaus and Yang, 1996) contains an iterative algorithm to find welfare weights that are consistent with zero income transfers between regions and equal *weighted* marginal utility. A consequence of the approach in RICE is that welfare weights depend on climate policies and vary between scenarios.

where we assume  $S_0^g = 0$ , and we require  $\lim_{t\to\infty} S_t^g/r_1^t = 0$ . Public savings remain zero,  $S_t^g = 0$ , if and only if the period-by-period balanced budget (16) holds. It is a matter of straight verification that private plus public savings equal investments,

$$K_{t+1} = N_t s_t + S_t^g. (22)$$

Furthermore, the relative value of future output depends on investments, through (12),

$$b_t = \frac{K_{t+1}}{\alpha Y_t}.$$
(23)

Equations (22) and (23) imply that the weight given to future output  $b_t$  corresponds oneto-one with private plus public savings. A planner who attributes high welfare weights to future generations transfers income from present generations to future generations, through positive public savings. The increased future capital stock also increases the value of future output relative to current output. In most of our analysis, we close down a preferential treatment of future generations by requiring a government balanced budget (16). A positive carbon price  $p_t > 0$  can be used to address climate externalities in a market equilibrium; imposing (16) rules out other income intervention objectives.

The next definition zooms in on optimal carbon policies.

**Definition 3.** A competitive equilibrium is called 'optimal' if carbon prices,  $p_t$ , and lump-sum transfers  $\tau_t$  are set such that the allocation maximizes welfare

$$W = \sum_{t=1}^{\infty} \beta_t N_t u_t, \tag{24}$$

for some welfare weights sequence  $\beta_t$ , satisfying  $\sum_{t=1}^{\infty} \beta_t N_t < \infty$ .

The above definition does not impose restrictions to the welfare weights sequence. Below we will see that in our economy, in a representative optimal equilibrium, welfare weights  $\beta_t$  depend on demography.<sup>18</sup> We note that all optimal equilibria are Paretoefficient.<sup>19</sup>

<sup>&</sup>lt;sup>18</sup>In a typical infinitely-lived representative-agent (Ramsey-Cass-Koopmans) model, we have  $\beta_t = \beta^t$  for some exogenous  $\beta$ .

<sup>&</sup>lt;sup>19</sup>We do not exclude the possibility that a Pareto-efficient allocation exists that is not optimal in the above sense, e.g. when  $\beta_t = 1$ , so that the weight series sum to infinity and welfare is not defined.

## 4. Analysis

#### 4.1. Existence

We prove existence for a representative equilibrium, and for the unique optimal representative equilibrium. We can then study the properties, specifically the effect of demography on carbon prices. The proofs are relegated to Appendix B. The propositions state both existence and properties for our OLG economy, which we compare with those for the RCK model in the next section.

The first proposition shows how the investment share varies with demography. We note that individual savings depend on both aging and fertility (eq (6)), but at the aggregate level, the savings rate only depends on life-expectancy.

**Proposition 1.** For any carbon pricing rule  $p_t = g_t Y_t$ , for given sequence  $g_t \ge 0$ , a unique representative competitive equilibrium exists. The investment shares depend on demography, but not on carbon policies:

$$\frac{K_{t+1}}{Y_t} = \frac{\delta h_t (1-\alpha)}{1+\delta h_t},\tag{25}$$

and so does the effective discount factor (19)

$$b_t^* = \frac{\delta h_t (1 - \alpha)}{(1 + \delta h_t)\alpha}.$$
(26)

*Proof.* In appendix **B**.

The representative equilibrium, as such, does not require efficient carbon policies; there is no match between the carbon price and the social costs of carbon. Below, we will study efficient policies in a representative equilibrium. To ensure bounded welfare weights, we make the following assumption.

**Assumption 3.** (Bounded net present value) Preference and production parameters satisfy  $\delta h_t(1-\alpha) < (1+\delta h_t)\alpha$ , for all t.

The second step is that we prove uniqueness of the optimal competitive equilibrium, for given welfare weights, and we show that in such an equilibrium the market costs of carbon and social costs of carbon coincide.

**Proposition 2.** For any weight sequence  $\beta_t$  such that  $\sum_{t=1}^{\infty} \beta_t N_t < \infty$ , a unique optimal competitive equilibrium exists. In this equilibrium, the carbon price equals the market costs of carbon,  $p_t = MCC_t$ . Carbon prices are proportional to income, and increase with weights given to

future generations,

$$p_t^* = Y_t \sum_{i=0}^{\infty} \theta_i b_t^{t+i}, \tag{27}$$

where  $b_t^{t+i} = b_t \cdot b_{t+1}...b_{t+i}$ , and  $b_t$ , as in (19), is given by

$$b_t^* = \frac{\sum_{j=0}^{\infty} \alpha^j (\beta_{t+j+1} N_{t+j+1} + \beta_{t+j} \delta h_{t+j} N_{t+j})}{\sum_{j=0}^{\infty} \alpha^j (\beta_{t+j} N_{t+j} + \beta_{t+j-1} \delta h_{t+j-1} N_{t+j-1})}.$$
(28)

*Proof.* In appendix **B**.

We have now shown existence of a unique representative equilibrium given a carbon pricing rule (Prop 1), and existence of a unique optimal equilibrium given welfare weights (Prop 2). The next proposition connects the two, proving existence of a unique optimal representative equilibrium.

**Proposition 3.** *A unique optimal representative competitive equilibrium exists, with welfare weights dependent on demography, but independent of climate parameters,* 

$$\beta_{t+1}^* = n_t^{-1} \frac{\delta h_t (1-\alpha)}{(1+\delta h_t)\alpha} \beta_t.$$
(29)

*The relative value of future output (19) is given by (26); the efficient carbon price is given by* 

$$p_t^* = Y_t \sum_{i=1}^{\infty} \theta_i \prod_{j=0}^{i-1} \frac{\delta h_{t+j}(1-\alpha)}{(1+\delta h_{t+j})\alpha}.$$
(30)

*Proof.* In appendix **B**.

The proposition reveals various important features of the optimal representative equilibrium. The weights given to future generations increase with an aging population, not because aging increases the population size per se, but because aging changes the savings decisions, and thereby changes the weights given to future output (Prop 1), and indirectly the weights given to future consumption. Welfare weights do not depend on the climate characteristics, that is, welfare weights are independent of climate policies and can be taken as input.<sup>20</sup> That is, a laissez-faire equilibrium with no carbon prices,  $p_t = 0$ , and no income transfers  $\tau_t = 0$ , implements the same implicit welfare weights as the optimal representative carbon policy of Prop 3.

<sup>&</sup>lt;sup>20</sup>This is different from the RICE model; see also footnote 17.

#### 4.2. Comparative demography

Here we consider, for given technologies  $A_t(.)$  and  $\Omega(.)$  and preferences u(.), how changes in demographic parameters  $n_t$  and  $h_t$  affect equilibrium outcomes. Fertility is a determinant of future labour supply, and as such, of economic growth. Aging determines savings, and as such also affects future carbon prices. The next corollaries establish the effect of demography on output.

But before we study demographic change, the next corollary establishes the benchmark case; Golosov et al. (2014)'s result is replicated if we shut down demographic change:

**Corollary 1.** Without demographic change,  $n_t = 1$ ,  $h_t = h$ , optimal representative carbon prices are given by

$$p_t^* = Y_t \sum_{i=0}^{\infty} \theta_i \beta^i,$$

for constant  $\beta = \frac{\delta h(1-\alpha)}{(1+\delta h)\alpha}$ , so that carbon prices are proportional to output,  $\frac{p_t^*}{Y_t} = \frac{p_{t+1}^*}{Y_{t+1}}$ .

*Proof.* Follows immediately from Proposition 3, equations (29) and (30).

Then we establish two results of demography for capital and output, along the lines of Canton and Meijdam (1997). These corollaries follow from Proposition 1.

**Corollary 2.** *In a representative equilibrium, aging (keeping fertility unchanged) increases the future capital stock, increases output, and decreases the return to investments.* 

for 
$$i > 0$$
;  $\frac{dK_{t+i}}{dh_t} > 0$ ,  $\frac{dY_{t+i}}{dh_t} > 0$ .  $\frac{dr_{t+i}}{dh_t} < 0$ .

*Proof.* In appendix **B**.

**Corollary 3.** *In a representative equilibrium, decreasing fertility (keeping life-expectancy un-changed) increases the immediate next capital stock and decreases returns to investments.* 

$$\frac{dK_{t+1}}{dn_t} < 0, \ \frac{dr_{t+1}}{dn_t} > 0.$$

*Proof.* In appendix **B**.

The above two corollaries capture the spirit of the empirical literature, which suggests that both increasing life-expectancy and decreasing fertility over the last decades have increased aggregate household savings rates and decreased the capital returns (Imrohoroglu and Zhao, 2017; Carvalho et al., 2016; Curtis et al., 2015; Eggertsson and Mehrotra, 2014; Bloom et al., 2007, 2003; Curtis et al., 2015; Sánchez-Romero, 2013). The last corollary remains silent on the effect of fertility on output, and the long run. An increased labour supply compensates for the decrease in capital so that the immediate output effect is ambiguous. For the long run effects, we note that emissions can either rise or fall with changing labour supply. Thus, future production may increase through increased labour supply, but it can also decrease through climate damages.

The next corollaries take the model to the domain of climate policies. The effect of fertility is ambiguous, as in the previous corollary. But for life expectancy, we have clear results. We compare a reference economy with a hypothetical alternative that has the same history, the same technology, but some different demographic future. An increase in current or future life-expectancy increases carbon prices by increasing the value associated with future damages:

**Corollary 4.** *In an optimal representative competitive equilibrium, present carbon prices increase with present and future increases in life-expectancy* 

$$\frac{dp_t}{dh_{t+k}} > 0, \; \forall \; k \ge 0.$$

*Proof.* In appendix **B**.

In addition to an increase in the present carbon price, an increased life-expectancy also leads to a faster rise in carbon prices:

**Corollary 5.** Assume that life-expectancy is non-decreasing,  $h_t \le h_{t+1}$ , and has not reached its long-term level at time  $t^*$ ,  $h_{t^*} < h_{\infty}$ , then carbon prices in an optimal representative competitive equilibrium increase faster than income for all  $t < t^*$ :

$$\frac{p_{t+1}^*}{Y_{t+1}} > \frac{p_t^*}{Y_t}$$

*Proof.* In appendix **B**.

The general approach in the IAM literature is that parameters are calibrated on the basis of historically observed macro variables, and then the model is used to project forwards output, investment and consumption paths. The above propositions show that demographic change creates a wedge between future projections and past observations. This result will, later on, be contrasted with the effect of demography on projections in RCK models. We formalize the insights as a Remark, where we compare a model that abstracts from demography to a model including demographic change.

**Remark 1.** Consider two economies with same technology and labour supply, calibrated to the same historic observations (t = 0, 1) for economic growth and returns on capital,  $Y_0, Y_1, r_1$ . In the benchmark economy, life-expectancy is assumed stationary. In the extended economy (denoted by a tilde), life expectancy is increasing,  $\tilde{h}_t \leq \tilde{h}_{t+1}$ , and has not reached its long-term level at time  $t^*$ ,  $\tilde{h}_{t^*} < \tilde{h}_{\infty}$ . In both economies, preference parameter  $\delta, \tilde{\delta}$  is calibrated such that (25,26) hold in the calibrated period t = 0. Then optimal carbon prices are higher in the economy with dynamic demography. For all t

 $\widetilde{p}_t > p_t.$ 

## 5. The Ramsey-Cass-Koopmans model

This section presents the standard neoclassical growth model with an climate module to derive the conditions through which demography affects climate policies. We assume a logarithmic utility function, full depreciation and an identical production function as in the benchmark model. We also extend the analysis to the case where the social planner can use either a Benthamite or a Millian utility as in Canton and Meijdam (1997). The Benthamite approach suggests to maximize social welfare by weighting the utility function by the size of the dynasty  $M_t$ , while the Millian approach uses no weights. The parameter  $0 \le \xi \le 1$  allows us to identity between those extreme cases. To certain extent, we can relate the former case to the usual assumption in the integrated assessment models  $\xi = 1$  (see e.g., Nordhaus (2008)) and the latter to the modeling in Golosov et al. (2014), which implies  $\xi = 0$ .

#### 5.1. Households

A representative household derives utility from aggregate consumption  $C_t$ , while population size is given by  $M_t$ ,<sup>21</sup>

$$\max_{\{C_t, S_t \ge 0\}_{t=0}^{\infty}} \sum_{t=0}^{\infty} \beta^t M_t^{\xi} \ln \left( C_t / M_t \right),$$
(31)

subject to,

<sup>&</sup>lt;sup>21</sup>In terms of the OLG model, we have  $M_t = N_t + h_{t-1}N_{t-1}$ .

$$C_t + S_t \le r_t S_{t-1} + w_t L_t + M_t \tau_t, \quad \forall t.$$
(32)

The parameter  $\beta \in (0, 1)$  is the representative consumer's discount factor,  $\xi = 0$  refers to a Millian (average) welfare function, while  $\xi = 1$  refers to a Benthamite (aggregate) welfare function, labor supply  $L_t$  is inelastic and typically proportional to total population. Population growth is described through  $M_{t+1} = m_t M_t$ .

The government sets carbon prices and income transfers that maximize the representative household's welfare function. It is immediately clear that the distribution of income transfers over time has no effect on the consumption distribution by the households. For convenience, we consider the case where the government returns tax revenues at the same period, (16), then savings equal capital,  $S_t = K_t$  and the equilibrium equals the optimal solution to maximization of (31) subject to (7) and the commodity balance,

$$C_t + K_{t+1} = Y_t.$$

#### 5.2. Analysis

We first establish the general properties of the RCK model, both in terms of capital investments and in terms of carbon policies.

**Proposition 4.** In the RCK model with dynamic population  $M_t$ , the investment share depends on demography, but not on climate parameters,

$$\frac{K_{t+1}}{Y_t} = \frac{\sum_{j=1}^{\infty} \alpha^j \beta^j M_{t+j}^{\xi}}{\sum_{j=0}^{\infty} \alpha^j \beta^j M_{t+j}^{\xi}}.$$
(33)

The optimal carbon price equals the market cost of carbon and is proportional to income,

$$p_t^* = Y_t \sum_{i=1}^{\infty} \theta_i b_t^{t+i}, \tag{34}$$

where  $b_t^{t+i} = b \cdot b_{t+1} \dots b_{t+i}$ , and  $b_t$ , as in (19), is given by

$$b_t^* = \frac{\sum_{j=0}^{\infty} \alpha^j \beta^{j+1} M_{t+j+1}^{\xi}}{\sum_{j=0}^{\infty} \alpha^j \beta^j M_{t+j}^{\xi}}.$$
(35)

*Proof.* In appendix **B**.

The OLG and RCK economies produce similar carbon price formulas if we neglect demographic dynamics:

**Corollary 6.** For  $\xi = 0$ , or  $\xi > 0$  without demographic change,  $M_t = M$ , optimal representative carbon prices are given by

$$p_t^* = Y_t \sum_{i=0}^{\infty} \theta_i \beta^i,$$

and carbon prices are proportional to output,  $\frac{p_t}{Y_t} = \frac{p_{t+1}}{Y_{t+1}}$ .

*Proof.* In appendix **B**.

Now we consider the effects of demographic change. For the OLG model, we considered an increase in life-expectancy and fertility. The RCK model features population size, which is derived from life-expectancy and fertility. We first consider the effect on investments and output, which were established as the main factors determining the carbon price:

**Corollary 7.** In the RCK economy with Benthamite welfare  $\xi = 1$ , population growth (assuming the future population size  $M_t$  is derived from  $M_1$  and  $(m_2, ..., m_t)$ ) increases the future capital stock and output,

$$\forall i \ge 1 : \frac{dK_{t+i}}{dm_t} > 0, \frac{dY_{t+i}}{dm_t} > 0.$$

*Proof.* In appendix **B**.

The above corollary provides the intuition for the next one:

**Corollary 8.** For  $\xi > 0$ , present carbon prices increase with present and future increases in population size

$$\frac{dp_t}{dm_{t+i}} > 0 \text{ for } i \ge 0.$$

*Proof.* In appendix **B**.

Evaluating the corollaries, and comparing these with their equivalent corollaries for the OLG model (1,2,3,4,5), we see that both investments and carbon price dynamics respond very differently to demographic change in the RCK model. Whereas the global trends of increasing life-expectancy leads to rising savings and fast-rising carbon prices in the OLG model, in the RCK model, the decrease in population growth leads to a decrease in the investment share, and carbon prices that increase less then with output. **Corollary 9.** Assume  $\xi > 0$  and population growth is non-increasing,  $m_t \ge m_{t+1}$ , and has not reached its long-term level,  $m_t > m_{\infty}$ , then carbon prices increase less than income

$$\frac{p_{t+1}}{Y_{t+1}} < \frac{p_t}{Y_t}.$$

*Proof.* In appendix **B**.

Comparing the corollaries with their equivalent for the OLG economy, we find sharp analytical results. Demography has opposite impacts on the paths of optimal carbon prices, dependent on the model structure. In the OLG framework, a structural increase in life-expectancy leads to higher carbon prices that grow faster than the economy. In the RCK models the outcome depends on the welfare function. On the one hand, if we apply Millian welfare ( $\xi = 0$ ), carbon prices increase proportionally to output. On the other hand, assuming a decreasing population growth and a weighted utility function (the Benthamite case,  $\xi = 1$ ), it turns out that carbon prices increase less than with income.

The next remark presents the effect of demographic change on calculated optimal climate policy.

**Remark 2.** Consider two economies with same technology and labour supply, calibrated to the same historic observations (t = 0, 1) for economic growth and returns on capital,  $Y_0, Y_1, r_1$ . In the benchmark economy, population growth is assumed constant or absent, or welfare is Millian  $(\xi = 0)$ . In the extended economy (denoted by a tilde), population growth decreases,  $\tilde{m}_t \ge \tilde{m}_{t+1}$ , and has not reached its long-term level at  $t = t^*$ :  $\tilde{m}_{t^*} > \tilde{m}_{\infty}$ . In both economies, preference parameter  $\beta, \tilde{\beta}$  is calibrated such that (33) holds in the calibrated period t = 0. Then optimal carbon prices are lower in the economy with dynamic demography. For all  $t \le t^*$ ,

$$\widetilde{p}_t < p_t.$$

Taking the two remarks together, we can now compare the effects of demographic change in the OLG model versus the RCK model. Both models, when calibrated to the same historic output levels and returns on capital, and when abstracting from demography, will present the same optimal carbon prices. But results radically change if both models are calibrated to the same future demographic change of increasing life-expectancy, decreasing fertility, increasing population, and decreasing population growth. Then both models, when used to project future trajectories, will produce con-

trasting efficient carbon prices. The OLG economy will present high carbon prices (comparatively), that increase faster than output. The RCK model will present lower carbon prices, that increase at a rate below the growth of the economy. To assess the quantitative substance of this analytical result, we add a numerical calibration in the next section.

## 6. Quantitative assessment

In this section, we evaluate quantitatively carbon prices paths for the next hundred years, using our benchmark OLG model and the RCK models, with and without population dynamics. A key characteristic of our assessment is that, though the model is defined for periods that span many years, we set up our calibration and simulation such that all variables are calculated on an annual basis. First, we specify the calibration of annual values for life expectancy  $h_t$ , fertility  $n_t$  and population size and growth  $(M_t, m_t)$ . Then we describe the annual climate response function  $\theta_i$ . These parameters, and Golosov et al. (2014) as reference, are used to calibrate the preference parameter  $\delta$ . We subsequently calculate the welfare weights  $\beta_t$  and output weights  $b_t$  in our model, both expressed as annualized variables. Welfare weights  $\beta_t$  in the RCK model are exogenous, but we calculate the RCK models' output weights  $b_t$ , which depend on economic growth if the elasticity of marginal utility is not equal to one. Finally, we bring all the variables together to determine the social costs of carbon for the various models.

#### 6.1. Demographics calibration

Our benchmark model considers 30 years per period, excluding childhood, so that life time is maximally two periods. For low-income countries with low life-expectancy, we assume that consumers enter the labour force at age 15, so that 1 + h = (LE - 15)/30, where *LE* is life-expectancy. For high-income countries with high longevity, we assume that the average consumer starts working at age 20 as usual in previous literature, so that 1 + h = (LE - 20)/30. Assuming a smooth transition between the low and highincome countries, we calibrate our parameter  $h_t$  through

$$h_t = \min\{0.03LE_{t'} - 1.35, 1\},\$$

which we calculate for each year t. In our model,  $h_t$  refers to the life-expectancy of the currently young, age 15, while our database measures life-expectancy at birth. To match the model parameter with data, we define life-expectancy in our model at time t through life-expectancy in the database at t' = t - 15. We use the medium-variant projection of world average data that come from the United Nations World Population Prospects, 2015 revision. Projections are available in five-year time intervals, which we linearly interpolate for annual time series. We cap the model life-expectancy at  $h_t = 1$  for projected world average life-expectancies exceeding 78 years.

For fertility, the number of children per adult is half the births per women,  $FR_t$ ,

$$n_t = FR_t/2.$$

The historic estimates and projections come from the same UN database.

In contrast to the OLG model, demography in the RCK models only requires a measure for population size and growth,  $(M_t, m_t)$ , where  $M_t$  denotes total population (both sexes) at period t. We use the same medium-variant projections from the UN database.<sup>22</sup>

#### 6.2. Preferences calibration

In order to create time series for the discount factor  $b_t$ , in the calibration of the OLG model, we determine parameter  $\delta$  such that  $b_t$ , as defined in (26), equals the discount factor b = 0.985 used in Golosov et al. (2014) at 2010. The annualized value ratio associated with the year 2010 is then given by  $b_t^{1/30}$ . Notice that the parameter  $b_t$  is based on the life-expectancy at the year 1995. Thus, to calibrate our parameter  $\delta$ , we set  $\alpha = 0.3$ , life expectancy  $h_{2010} = 0.58$ , output discount factor  $b_{2010} = \beta^{GHKT} = 0.985^{30}$  and solve for  $\delta$  as follows:

$$\delta = \frac{\alpha b_{2010}}{h_{2010}[1 - \alpha - \alpha b_{2010}]} = 0.645$$

#### 6.3. Discount factor calculation

Using the above values for the capital share  $\alpha$ , preferences  $\delta$ , life expectancy  $h_t$ , fertility  $n_t$ , we simulate the time series for  $b_t$  from (26).

<sup>&</sup>lt;sup>22</sup>Due to high uncertainty about long-run population prospects, Rozell (2017) recommends to use the International Institute for Applied Systems Analysis (IIASA) population projections in climate policy studies since that dataset implies more reliable demographic scenarios. Our calibration relies on the United Nations estimations for ease of comparison with previous literature.

For the RCK model, we apply (35) to calculate the output ratio  $b_t$  for those models with logarithmic utility,  $\eta = 1$ . The sequence  $b_t$  only depends on intergenerational preferences  $\beta$  and population growth rates  $m_t$ . For the general model with constant elasticity of substitution  $\eta \neq 1$ , as in Weitzman (2007); Nordhaus (2008, 2014), we conjecture that the output ratio is approximated by

$$b_{t} = \frac{\sum_{j=0}^{\infty} \alpha^{j} \beta^{j+1} (\gamma_{t}^{t+j+1})^{1-\eta} M_{t+j+1}^{\xi}}{\sum_{j=0}^{\infty} \alpha^{j} \beta^{j} (\gamma_{t}^{t+j})^{1-\eta} M_{t+j}^{\xi}}$$
(36)

where  $\gamma_t^{t+j}$  is the compounded productivity growth between period t and t + j. The historic estimates for productivity growth are based on World Bank data and the SSP scenarios (see Appendix A for more details). For future projections, we assume a constant annual increase of per capita income of 2 per cent per year. The formula collapses to (35) for  $\eta = 1$ . We assume stationary population,  $n_t = m_t = 1$  and constant  $\gamma_t$  after 2100, and calculate the output ratios (36) backwards, as specified in detail in Appendix E. We note that, in balanced growth, the output ratio formula becomes  $b = \beta \gamma^{1-\eta} m^{\xi}$ . In appendix C, we show that, indeed, this formula is correct for a balanced growth path. As a static approximation, we use the formula presented in the introduction to check the numbers presented in Table 1

$$b_t \approx \beta m_t^{\xi} \gamma_t^{1-\eta}. \tag{37}$$

Figure 2 depicts the effective annual discount factors, calculated recursively, of our benchmark model. To relate our approximation to previous analysis, it also displays the discount factors used in Golosov et al. (2014), which abstracts from demography, and other RCK models with a Benthamite welfare function. Notice that to calculate the productivity growth factor  $\gamma_t$  during the period 1970-2015, we use 10-year average of GDP per capita growth rates to smooth the business cycle features of the data.

The outstanding feature of the figure is the contrasting patterns between our OLG model and the RCK models with Benthamite welfare. In our model, global demographic patterns, specially the worldwide rise of life-expectancy, lead to a rise of the value of future output  $b_t$ . In contrast, the RCK models show a discount factor that decreases over time due to declining population growth rates. As we will see shortly, these divergent patterns for the discount factor have a significant impact on carbon prices.<sup>23</sup>

<sup>&</sup>lt;sup>23</sup>Notice also that with constant life-expectancy h and  $n_t = 1$ , our model would match the discount factor in Golosov et al. (2014).



Figure 2: Effective annual discount factor  $b_t$ , 1970-2100

Note: The year 2017 is indicated with a vertical line.  $b_t$  is calculated as in equation (26). The parameter  $\delta$  is calibrated such that in 2010, the value of  $b_{2010}$  corresponds to 0.985, as in Golosov et al. (2014). The discount factors for the RCK models are given by (36), see Table 1 for more details and parameter values.

#### 6.4. Climate dynamics calibration

We use a stylized climate module as in van den Bijgaart et al. (2016) and Gerlagh and Liski (2017b). The dynamics of damages  $\theta_i$  are governed by carbon exchange between reservoirs, which is conveniently described through an exponential CO<sub>2</sub> decay function. Damages follow temperature, which slowly converges to its equilibrium level for given atmospheric CO<sub>2</sub>; a higher atmospheric CO<sub>2</sub> content leads to slowly increasing damages. The reduced form ((Gerlagh and Liski, 2017b), Theorem 1), is given by a 'multi-box' representation,

$$\theta_i = \sum_j \sum_k a_j b_k \pi \varepsilon_k \frac{(1 - \eta_j)^i - (1 - \varepsilon_k)^i}{\varepsilon_k - \eta_j},\tag{38}$$

where  $\eta_j$  are the atmospheric depreciation rates,  $\varepsilon_k$  are the temperature adjustment speeds,  $a_j$  and  $b_k$  are the shares of the relevant processes, and  $\pi$  is the long-run emissionsdamages sensitivity. We take the parameters from van den Bijgaart et al. (2016), Table 6 and 7, for the median carbon model,  $a = (0.220, 0.279, 0.278, 0.222), \eta = (0, 0.0035, 0.0507, 0.2892),$ and temperature models,  $b = (0.2218, 0.3306, 0.4476), \varepsilon = (0.9787, 0.1980, 0.0036), \pi =$ 0.0167. These parameters indicate that 22 per cent of emissions remain in the atmosphere 'forever', while the same share very quickly transits to other carbon reservoirs (at about 30% per year). Also, about 22% of temperature adjustment is virtually immediate, while almost half of temperature adjustment only happens after more than a century.

#### 6.5. The social cost of carbon

We have now everything in place to determine the social costs of carbon. The carbon price factors  $g_t$ , as defined by (18), are calculated backwards from 2100 to 2000; the procedures are described in the Appendices D and E. We multiply the values for  $g_t$  by output  $Y_t$  to quantify the social costs of carbon.

Figure 3 displays the social cost of carbon for the models considered. The implied carbon price at 2017 based on the model structure of Golosov et al. (2014) is about 25  $\notin/tCO_2$ , growing in-step with the size of the economy. Our model calculates a carbon price rising faster than the economy, but also at a much higher level, amounting to about 85  $\notin/tCO_2$ . The RCK models with Benthamite utility, on the other hand, have carbon prices that grow less than with income, *and* start at a lower level. Notice that a worldwide rising carbon price starting at 85  $\notin/tCO_2$  exceeds by magnitude the objectives stated in the Paris Intended Nationally Determined Contributions (INDCs).



Figure 3: The social cost of carbon, 2000-2100

Note: The year 2017 is indicated with a vertical line.

These results suggest that demographic patterns are an important factor in the calculation of the social costs of carbon. As shown in Table 1, the carbon price in our OLG model at 2100 attains a value of approximately 800 EUR/tC02, a price that is about four times higher than the carbon price derived from a model with no demography as in Golosov et al. (2014).

## 7. Discussion

In this paper we explore optimal environmental policies in a climate-economy with population dynamics that entail changes in savings patterns and capital returns. Using a stylized overlapping generations model, we show that an increase in life-expectancy and a decrease in fertility rates raise households' savings, declining the rate of return to capital due to its relative abundance. Changing savings patterns also lead to changes in the way people value future economic gains or losses, including climate damages. We propose a time-varying effective discount factor that allows climate policies to reflect demographic patterns.

Our results show that demographic factors in climate-economy models have an important effect on climate policies. We developed a stylized OLG model and Ramsey-Cass-Koopmans model with life-expectancy and fertility features. For both types of models we compared one with and one without population dynamics, assuming that they are calibrated to historic savings. We obtain outstanding divergent patterns for optimal carbon prices between OLG and RCK models. The OLG and RCK models without demography replicate the same carbon price projections. But, when we consider the expected global demographic trends as in Figure 1, we find much higher carbon prices, that also grow faster than the economy, for the OLG model, while we find lower carbon prices, that also grow below the rate of economic growth, for the RCK models.

Our study thus points to the importance of testing the long-term empirical validity of integrated assessment models, in terms of the structural connection between demography and macro-economic savings and investments, when used for climate policy advice. The OLG and RCK models, both with rigorous micro-foundation, give radically different answers, and as our quantitative assessment shows, may lead to an under- or over-reporting of the social costs of carbon by large amount.

### References

Backus, D., Cooley, T., and Henriksen, E. (2014). Demography and low-frequency capital flows. *Journal of International Economics*, 92:S90–S102.

- Balestra, C. and Dottori, D. (2012). Aging society, health and the environment. *Journal of Population Economics*, 25:1045–1076.
- Barrage, L. (2014). Sensitivity analysis for golosov, hassler, krusell, and tsyvinski (2014): Optimal taxes on fossil fuel in general equilibrium. *Econometrica Supplementary Material*, 82(1):41–88.
- Barrage, L. (2016). Optimal dynamic carbon taxes in a climate-economy model with distortionary fiscal policy. Working paper.
- Becker, G. S. and Lewis, H. G. (1973). On the interaction between the quantity and quality of children. *Journal of political Economy*, 81(2, Part 2):S279–S288.
- Belfiori, E. (2013). Carbon taxes, altruism and the time consistency of climate policies.
- Bloom, D. E., Canning, D., and Graham, B. (2003). Longevity and life-cycle savings. *The Scandinavian Journal of Economics*, 105(3):319–338.
- Bloom, D. E., Canning, D., Mansfield, R. K., and Moore, M. (2007). Demographic patterns and household saving in china. *Journal of Monetary Economics*, 54(1):91–114.
- Bohn, H. and Stuart, C. (2015). Calculation of a population externality. *American Economic Journal: Economic Policy*, 7(2):61–87.
- Canton, E. and Meijdam, L. (1997). Altruism and the macroeconomic effects of demographic changes. *Journal of Population Economics*, 10:317–334.
- Carvalho, C., Ferrero, A., and Nechio, F. (2016). Demographics and real interest rates: Inspecting the mechanism. *European Economic Review*, 88:208–226.
- Casey, G. and Galor, O. (2017). Is faster economic growth compatible with reductions in carbon emissions? the role of diminished population growth. *Environmental Research Letters*, 12.
- Chen, K. and Wen, Y. (2017). The great housing boom in china. *American Economic Journal: Macroeconomics*. forthcoming.
- Cipriani, G. P. (2014). Population aging and payg pensions in the olg model. *Journal of Population Economics*, 27:251–256.

- Curtis, C. C., Lugauer, S., and Mark, N. C. (2015). Demographic patterns and household saving in china. *American Economic Journal: Macroeconomics*, 7(2):58–94.
- Dalton, M., O'Niel, B., Prskawetz, A., Jiang, L., and Pitkin, J. (2008). Population aging and future carbon emissions in the united states. *Energy Economics*, 30:642–675.
- De la Croix, D. and Gosseries, A. (2012). The natalist bias of pollution control. *Journal of Environmental Economics and Management*, 63(2):271–287.
- Eggertsson, G. B. and Mehrotra, N. R. (2014). A model of secular stagnation. NBER Working Paper No. 20574.
- Fanti, L. and Gori, L. (2012). Fertility and payg pensions in the overlapping generations model. *Journal of Population Economics*, 25:955–961.
- Fedotenkov, I., van Groezen, B., and Meijdam, L. (2014). Demographic change, international trade and capital flows. *Open Economies Review*, 25(5):865–883.
- Gerlagh, R. and Liski, M. (2017a). Carbon prices for the next hundred years. *The Economic Journal*. forthcoming.
- Gerlagh, R. and Liski, M. (2017b). Consistent climate policies. *Journal of the European Economic Association*. forthcoming.
- Gerlagh, R. and van der Zwaan, B. (2001). The effects of ageing and an environmental trust fund in an overlapping generations model on carbon emission reductions. *Ecological Economics*, 36:311–326.
- Golosov, M., Hassler, J., Krusell, P., and Tsyvinski, A. (2014). Optimal taxes on fossil fuel in general equilibrium. *Econometrica*, 82(1):41–88.
- Guo, J., Hepburn, C. J., Tol, R. S., and Anthoff, D. (2006). Discounting and the social cost of carbon: a closer look at uncertainty. *Environmental Science & Policy*, 9(3):205–216.
- Harford, J. D. (1998). The ultimate externality. *The American Economic Review*, 88(1):260–265.
- Howarth, R. B. (1998). An overlapping generations model of climate-economy interactions. *The Scandinavian Journal of Economics*, 100(3):575–591.

- Imrohoroglu, A. and Zhao, K. (2017). The chinese saving rate: Long-term care riks, family insurance and demographics. Working paper.
- Iverson, T. (2012). Optimal carbon taxes with non-constant time preference.
- Jouvet, P.-A., Pestieau, P., and Ponthiere, G. (2010). Longevity and environmental quality in an olg model. *Journal of Economics*, 100:191–216.
- Karp, L. and Rezai, A. (2014). The political economy of environmental policy with overlapping generations. *International Economic Review*, 55(3):711 733.
- Karp, L. and Traeger, C. (2013). Discounting. In Shogren, J. F., editor, *Encyclopedia* of Energy, Natural Resource, and Environmental Economics, volume 2, pages 286–292. Amsterdam: Elsevier.
- Krueger, D. and Ludwig, A. (2007). On the consequences of demographic change for rates of returns to capital and the distribution of wealth and welfare. *Journal of Monetary Economics*, 54(1):49–87.
- Leimbach, M., Kriegler, E., Roming, N., and Schwanitz, J. (2017). Future growth patterns of world regions – a {GDP} scenario approach. *Global Environmental Change*, 42:215 – 225.
- Mariani, F., Pérez-Barahona, A., and Raffin, N. (2010). Life expectancy and the environment. *Journal of Economic Dynamics & Control*, 34:798–815.
- Mastrandrea, M. D. and Schneider, S. H. (2001). Integrated assessment of abrupt climatic changes. *Climate Policy*, 1(4):433–449.
- Murtaugh, P. A. and Schlax, M. G. (2009). Reproduction and the carbon legacies of individuals. *Global Environmental Change*, 19(1):14–20.
- Nordhaus, W. (1993). Rolling the dice: An optimal transition path for controlling greenhouse gases. *Resource and Energy Economics*, 15:27–50.
- Nordhaus, W. (2008). *A question of balance: weighing the options on global warming policies*. New Haven, CT: Yale University Press.
- Nordhaus, W. (2014). Estimates of the social cost of carbon: Concepts and results from the dice-2013r model and alternative approaches. *Journal of the Association of Environmental and Resource Economists*, 1(1/2):273–312.

- Nordhaus, W. D. and Yang, Z. (1996). A regional dynamic general-equilibrium model of alternative climate-change strategies. *The American Economic Review*, pages 741–765.
- O'Neill, B. C., Kriegler, E., Riahi, K., Ebi, K. L., Hallegatte, S., Carter, T. R., Mathur, R., and van Vuuren, D. P. (2014). A new scenario framework for climate change research: the concept of shared socioeconomic pathways. *Climatic Change*, 122(3):387–400.
- Ono, T. (2005). The political economy of environmental taxes with an aging population. *Environmental & Resource Economics*, 30:165–194.
- Quaas, M. and Bröcker, J. (2016). Substitutability and the social cost of carbon in a solvable growth model with irreversible climate change. Working paper.
- Rezai, A. and van der Ploeg, F. (2015). Robustness of a simple rule for the social cost of carbon. *Economics Letters*, 132:48–55.
- Rozell, D. (2017). Using population projections in climate change analysis. *Climatic Change*, 142:521–529.
- Sánchez-Romero, M. (2013). The role of demography on per capita output growth and saving rates. *Journal of Population Economics*, 26(4):1347–1377.
- Stern, N. (2007). *The economics of climate change: The Stern review*. Cambridge, U.K.: Cambridge University Press.
- van den Bergh, J. and Botzen, W. (2015). Monetary valuation of the social cost of CO<sub>2</sub> emissions: A critical survey. *Ecological Economics*, 114:33–46.
- van den Bijgaart, I., Gerlagh, R., and Liski, M. (2016). A simple formula for the social cost of carbon. *Journal of Environmental Economics and Management*, 77:75–94.
- Varvarigos, D. (2010). Environmental degradation, longevity, and the dynamics of economic development. *Environmental & Resource Economics*, 46:59–73.
- Weitzman, M. L. (2007). A review of the stern review on the economics of climate change. *Journal of Economic Literature*, XLV:703–724.
- Weitzman, M. L. (2013). Tail-hedge discounting and the social cost of carbon. *Journal of Economic Literature*, 51(3):873–882.

## A. Data and Projections

The world GDP per capita data is obtained from 'World Development Indicators' of the World Bank. The GDP per capita at in constant 2010 US\$ for 1960 till 2015 is available in the aforementioned dataset. The projection of this variable beyond historical data and up to 2100 is done based on a SSP<sup>24</sup> scenario. We picked the second scenario, i.e. SSP2, which assumes moderate challenges in the future of environmental policies. Leimbach et al. (2017) predict the GDP per capita, under SSP2 scenario, to grow on an average rate of 2% per year from 2010 till 2100.

### **B.** Proofs

#### Lemma 1

*Proof.* We can rewrite the equality in intensive form as

$$g_t L_t a_t(e_t) = (1 - \alpha) a'_t(e_t).$$

where  $e_t = E_t/L_t$  is the emission intensity. The LHS is increasing in  $e_t$ , while the RHS is decreasing because of concavity. For  $e_t = 0$ , we have that the LHS is less than the RHS, for  $e_t = e_t^{\text{max}}$  we have that the LHS exceeds the RHS (which then equals zero). By continuity, there must be an  $e_t = E_t/L_t$  for which the RHS equals the LHS.

#### **Proposition 1**

*Proof.* We prove the proposition by induction, starting to establish an allocation and prices for t = 1, and then continuing for increasing t.

In the first period, the capital stock,  $K_1$  is known, labor supply  $L_1 = (1 - \chi n_1)$  is inelastic, and the emission price must satisfy  $p_1 = g_1Y_1$ . Using the firm's optimization on emission, (14), and Lemma 1, we establish a unique level of emissions of the firm in period one. Therefore, we have pinned down  $E_1$ , as well as  $L_1$  and  $K_1$ . The wage rate and the rate of interest in the first period are then determined by (13) and (12). The transfer in the first period is then

$$\tau_1 = \frac{p_1 E_1}{N_1}.$$

<sup>&</sup>lt;sup>24</sup>Shared Socio-economic Pathways are 5 scenarios for the path ahead of the world economy. These scenarios differ in terms of socioeconomic challenges for mitigation and adaptation activities. The scenarios are discussed in O'Neill et al. (2014).

and income of the young equals  $(1 - \alpha)Y_t$ . Because of logarithmic utility, a fixed share is consumed when young, the remainder saved.

$$N_1 c_{1,1} = \frac{1 - \alpha}{1 + \delta h_1} Y_t.$$
 (B.1)

The consumption of the old in the first period equals the value of capital

$$N_0 h_0 c_{2,1} = \alpha Y_t. \tag{B.2}$$

It follows immediately from the aggregate goods market equilibrium (20 that the remainder of output is invested. This establishes (25). Finally, (26) follows then from (12).

Now, assume that the economy is in equilibrium in period  $t \ge 1$ . We can invoke exactly the same equations to prove that it is in equilibrium in period t + 1 as well. Q.E.D.

#### **Proposition 2**

*Proof.* The condition  $\sum_{t=1}^{\infty} \beta_t N_t < \infty$  ensures that welfare and parameters  $b_t^{t+i}$  are well defined. To prove existence, we first solve the central planner's problem (24) given technology, (7, 8, 10), subject to the aggregate goods market clearing condition, (20). We then show that we can decentralize the optimal allocation as a competitive equilibrium.

The associated Lagrangian for the central planner's problem reads:

$$\mathcal{L} = \beta_0 \delta h_0 N_0 u(c_{2,1}) + \sum_{t=1}^{\infty} [\beta_t \{ N_t u(c_{1,t}) + \delta h_t N_t u(c_{2,t+1}) \} + \lambda_t \{ Y_t - N_t c_{1,t} - h_{t-1} N_{t-1} c_{2,t} - K_{t+1} \}]$$
(B.3)

where  $\lambda_t$  is the shadow price for the aggregate goods market clearing condition. We consider the optimal allocation  $c_{1,t}^*, c_{2,t}^*, Y_t^*, E_t^*, K_{t+1}^*$  and supporting dual variables  $\lambda_t^*$ , and show that we can construct prices  $r_t^*, w_t^*, p_t^*$ , savings  $s_t^*$  and transfers  $\tau_t^*$  such that all conditions (3-5), (12-14), (15), hold.

We construct prices to support the competitive equilibrium, through

$$\begin{aligned} r_t^* &\equiv \quad \frac{\partial Y_t}{\partial K_t} = \alpha \frac{Y_t^*}{K_t^*}, \\ w_t^* &\equiv \quad \frac{\partial Y_t}{\partial L_t}, \\ p_t^* &\equiv \quad \frac{\partial Y_t}{\partial F_t} \frac{\partial F_t}{\partial E_t}, \end{aligned}$$

so that FOCs (12-14) are satisifed by construction. The Lagrangean first-order conditions are (for  $c_{1,t}, c_{2,t}, K_{t+1}, E_t$  respectively):

$$\frac{\beta_t}{c_{1,t}^*} = \lambda_t^*,\tag{B.4}$$

$$\frac{\delta\beta_{t-1}}{c_{2,t}^*} = \lambda_t^*,\tag{B.5}$$

$$r_{t+1}^* = \frac{\lambda_t^*}{\lambda_{t+1}^*},\tag{B.6}$$

$$p_t^* = \sum_{i=0}^{\infty} \theta_i \frac{\lambda_{t+i}^*}{\lambda_t^*} Y_{t+i}^* \tag{B.7}$$

The first three FOCs give the households FOC (5). That is, all FOCs the define the competitive equilibrium are satisfied.

The savings  $s_t^*$  can subsequently be constructed from  $r_{t+1}^*$  and  $c_{2,t+1}^*$  and (4). The transfers  $\tau_t^*$  can subsequently be constructed from  $r_{t+1}^*$  and  $c_{1,t+1}^*$ ,  $w_t^*$ ,  $s_t^*$ , and (3). The regulator's budget (15) follows from Walras' law. Thus, we have shown that the optimal allocation is a competitive equilibrium. The last two FOCs (B.6, B.7) confirm that the optimal carbon price equals the market costs of carbon.

We now want to prove the analytical solution for the carbon prices. We guess and verify that the optimal allocation satisfies

$$\lambda_t^* Y_t^* = \sum_{j=0}^{\infty} \alpha^j \beta_{t+j} N_{t+j} + \alpha^j \beta_{t+j-1} \delta h_{t+j-1} N_{t+j-1}.$$
 (B.8)

We will see that this condition determines specific consumption shares to satisfy (B.4, B.5). Then, we find the investment shares, and show that this is consistent with (B.6). Finally, the carbon price rule (B.7) will be shown to produce (27).

The first two FOCs (B.4, B.5), together with our guess (B.8) imply consumption and investment shares that depend on demography, but are independent of the climate parameters:

$$\frac{N_t c_{1,t}^*}{Y_t^*} = \frac{\beta_t N_t}{\sum_{j=0}^{\infty} \alpha^j (\beta_{t+j} N_{t+j} + \beta_{t+j-1} \delta h_{t+j-1} N_{t+j-1})},$$
(B.9)

$$\frac{h_{t-1}N_{t-1}c_{2,t}^*}{Y_t^*} = \frac{\beta_{t-1}\delta h_{t-1}N_{t-1}}{\sum_{j=0}^{\infty} \alpha^j (\beta_{t+j}N_{t+j} + \beta_{t+j-1}\delta h_{t+j-1}N_{t+j-1})},$$
 (B.10)

The market clearing (20) subsequently defines the investment share,

$$\frac{K_{t+1}^*}{Y_t^*} = \frac{\sum_{j=1}^{\infty} \alpha^j (\beta_{t+j} N_{t+j} + \beta_{t+j-1} \delta h_{t+j-1} N_{t+j-1})}{\sum_{j=0}^{\infty} \alpha^j (\beta_{t+j} N_{t+j} + \beta_{t+j-1} \delta h_{t+j-1} N_{t+j-1})}.$$
(B.11)

And indeed, we see that if we substitute (B.8) into the dynamic FOC (B.6), we find the same expression for the investment share. Thus, the dual variables  $\lambda_t^*$  defined by (B.8) implements the optimal allocation  $c_{1,t}^*, c_{2,t}^*, Y_t^*, K_{t+1}^*$ .

To conclude, we immediately see that (B.7) and (B.8) result in the carbon price (27) mentioned in the proposition.

#### **Proposition 3**

*Proof.* We first assume that (29) holds, and show that  $\sum_{t=1}^{\infty} N_t \beta_t < \infty$ . Note that the recursive welfare weights rule implies

$$\beta_{t+1}N_{t+1} = \frac{\delta h_t(1-\alpha)}{(1+\delta h_t)\alpha}\beta_t N_t.$$

Assumption 3 thus guarantees that the sequence  $\beta_t N_t$  decreases expontentially, and the infinite sum of a geometric series is bounded. Therefore, welfare can be maximized.

Next, we note that for the welfare weights defined by (29), we can invoke Proposition 2 and its proof. The associated Lagrangian and its FOCs are exactly the same as (B.3) and the respective FOCs in the proof of proposition 2. Combining the first two FOCs with (B.1) and (B.2) for the representative equilibrium case, we get:

$$\frac{\beta_t}{\beta_{t-1}} = \delta \frac{c_{1,t}}{c_{2,t}} = n_{t-1}^{-1} \frac{\delta(1-\alpha)h_{t-1}}{(1+\delta h_{t-1})\alpha}$$

Therefore, the weights must satisfy (29).

For the carbon price, Proposition 2 tells us that we can use (18), in which we substitute (26) to arrive at (30).

#### **Corollary 2**

*Proof.* We prove this corollary by induction. For i = 1, we take the derivative for (25) with respect to  $h_t$ ; we have

$$\frac{dK_{t+1}}{dh_t} = \frac{\delta h_t (1-\alpha)}{(1+\delta h_t)^2} Y_t > 0.$$

Note that in a representative economy with  $g_t$  given, because of Lemma 1,  $E_{t+1}^*$  is independent of  $K_{t+1}$ , so that a rise in  $K_{t+1}$ , leads to an increased output (8). For  $r_{t+1}$ , using (12), we have

$$r_{t+1} = \alpha \Omega(z_{t+1}) K_{t+1}^{\alpha - 1} [A(E_{t+1}, L_{t+1})]^{1 - \alpha}.$$
(B.12)

According to lemma 1,  $E_{t+1}$  and  $L_{t+1}$  are not affected. Therefore, we have

$$\operatorname{sign}(\frac{dr_{t+1}}{dh_t}) = -\operatorname{sign}(\frac{dK_{t+1}}{dh_t}).$$

The corollary is thus correct for i = 1. For all future periods i > 1, we use forward induction. From (25), we get

$$\frac{dK_{t+i+1}}{dY_{t+i}} = \frac{\delta(1-\alpha)}{1+\delta h_t} > 0.$$

which subsequently leads to a rise in output  $Y_{t+i+1}$ , etc. Similar to above, we have

$$\operatorname{sign}(\frac{dr_{t+i+1}}{dh_t}) = -\operatorname{sign}(\frac{dK_{t+i+1}}{dh_t}).$$

#### Corollary 3

*Proof.* For capital stock of the next period, from (25), we have

$$\frac{dK_{t+1}}{dn_t} = \frac{\delta h_t (1-\alpha)}{1+\delta h_t} (1-\alpha) \frac{Y_t}{A(E_t, L_t)} [A_L(E_t, L_t) + A_E(E_t, L_t) \frac{dE_t}{dL_t}] (-\chi N_t) < 0,$$

or

$$\frac{dK_{t+1}}{dn_t} = \frac{\delta h_t (1-\alpha)}{1+\delta h_t} [w_t + p_t \frac{dE_t}{dL_t}](-\chi N_t) < 0.$$

For the rate of interest, using (B.12), we have

$$\frac{dr_{t+1}}{dn_t} = \frac{r_{t+1}}{A(E_{t+1}, L_{t+1})} [A_E(E_{t+1}, L_{t+1}) \frac{dE_{t+1}}{dL_{t+1}} + A_L(E_{t+1}, L_{t+1})](1 - \chi n_{t+1})N_t > 0.$$

For  $Y_{t+1}$ , taking the derivative with respect to  $n_t$ , we get

$$\frac{dY_{t+1}}{dn_t} = \frac{dK_{t+1}}{dn_t} \left[ \frac{dY_{t+1}}{dK_{t+1}} + \frac{dY_{t+1}}{dE_{t+1}} \frac{dE_{t+1}}{dK_{t+1}} \right] + \left(1 - \chi n_{t+1}\right) N_t \left[ \frac{dY_{t+1}}{dL_{t+1}} + \frac{dY_{t+1}}{dE_{t+1}} \frac{dE_{t+1}}{dL_{t+1}} \right],$$

or

$$\frac{dY_{t+1}}{dn_t} = \frac{dK_{t+1}}{dn_t} [r_{t+1} + p_{t+1}\frac{dE_{t+1}}{dK_{t+1}}] + (1 - \chi n_{t+1})N_t [w_{t+1} + p_{t+1}\frac{dE_{t+1}}{dL_{t+1}}].$$

The first term on the RHS is negative and the second term is positive.  $dY_{t+1}/dn_t > 0$  is then equivalent to

$$\frac{\delta h_t(1-\alpha)}{1+\delta h_t} [w_t + p_t \frac{dE_t}{dL_t}] [r_{t+1} + p_{t+1} \frac{dE_{t+1}}{dK_{t+1}}] \chi < (1-\chi n_{t+1}) [w_{t+1} + p_{t+1} \frac{dE_{t+1}}{dL_{t+1}}].$$

Using (25), a rise in output in t + 1 leads to an increase in  $K_{t+2}$  and, therefore, decreases  $r_{t+2}$ . Thereby, the output and the capital stock, as of t + 2, increases as well while the rates of interest decrease.

#### **Corollary 4**

*Proof.* Taking the derivative of (30) with respect to  $h_{t+k}$  for some  $k \ge 0$ , we get

$$\frac{dp_t^*}{dh_{t+k}} = Y_t \sum_{i=1}^{\infty} \theta_i \prod_{j=1}^{k-1} \frac{\delta h_{t+j}(1-\alpha)}{(1+\delta h_{t+j})\alpha} \prod_{j=k+1}^{i-1} \frac{\delta h_{t+j}(1-\alpha)}{(1+\delta h_{t+j})\alpha} \frac{\delta(1-\alpha)}{(1+\delta h_{t+k})^2\alpha} > 0.$$

Corollary 5

*Proof.* Using equation (30), we can write

$$\begin{aligned} \frac{p_{t+1}^{*}}{Y_{t+1}} &- \frac{p_{t}^{*}}{Y_{t}} = \\ \theta_{1} [\frac{\delta h_{t+1}(1-\alpha)}{(1+\delta h_{t+1})\alpha} - \frac{\delta h_{t}(1-\alpha)}{(1+\delta h_{t})\alpha}] + \theta_{2} \frac{\delta h_{t+1}(1-\alpha)}{(1+\delta h_{t+1})\alpha} [\frac{\delta h_{t+2}(1-\alpha)}{(1+\delta h_{t+2})\alpha} - \frac{\delta h_{t}(1-\alpha)}{(1+\delta h_{t+2})\alpha}] + \\ \theta_{3} \frac{\delta h_{t+1}(1-\alpha)}{(1+\delta h_{t+1})\alpha} \frac{\delta h_{t+2}(1-\alpha)}{(1+\delta h_{t+2})\alpha} [\frac{\delta h_{t+3}(1-\alpha)}{(1+\delta h_{t+3})\alpha} - \frac{\delta h_{t}(1-\alpha)}{(1+\delta h_{t})\alpha}] + \dots = \\ \frac{(1+\delta h_{t})\alpha}{\delta h_{t}(1-\alpha)} \sum_{i=1}^{\infty} \theta_{i} \prod_{j=1}^{i} \frac{\delta h_{t+j-1}(1-\alpha)}{(1+\delta h_{t+j-1})\alpha} \cdot [\frac{\delta h_{t+i}(1-\alpha)}{(1+\delta h_{t+i})\alpha} - \frac{\delta h_{t}(1-\alpha)}{(1+\delta h_{t})\alpha}]. \end{aligned}$$
(B.13)

Since

$$\partial \frac{\delta h_t (1-\alpha)}{(1+\delta h_t)\alpha} / \partial h_t = \frac{\delta^2 (1-\alpha)}{(1+\delta h_t)^2 \alpha} > 0,$$

and  $h_t$  is not decreasing in t, the last term in (B.13) is not negative, and as  $h_{t^*} < h_{\infty}$ , there is at least one future t with  $h_{t+1} < h_t$ , and one term is strictly positive. Therefore, we have

$$\frac{p_{t+1}^*}{Y_{t+1}} > \frac{p_t^*}{Y_t}$$

#### **Proposition 4**

*Proof.* The Lagrangian function in the government's problem is given by:

$$\mathcal{L}(\cdot) = \sum_{t=0}^{\infty} \beta^t M_t^{\xi} \ln\left(\frac{C_t}{M_t}\right) + \lambda_t \left[Y_t - C_t - K_{t+1}\right]$$
(B.14)

Following the same procedure as in the proof of proposition 2, and assuming that there is a unique solution for the competitive equilibrium, the first-order conditions with respect to  $C_t$ ,  $K_{t+1}$  and  $E_t$  can be written as follows:

$$\beta^t \frac{M_t^{\xi}}{C_t^*} = \lambda_t^* \tag{B.15}$$

$$\frac{\alpha Y_{t+1}^*}{K_{t+1}^*} = r_{t+1}^* = \frac{\lambda_t^*}{\lambda_{t+1}^*}$$
(B.16)

$$p_t^* = \sum_{t=1}^{\infty} \frac{\lambda_{t+i}^*}{\lambda_t^*} \theta_i Y_{t+i}^* \tag{B.17}$$

Likewise, we guess and verify that the sequence of optimal allocations satisfy:

$$\lambda_t^* Y_t^* = \sum_{j=0}^\infty \alpha^j \beta^{t+j} M_{t+j}^{\xi}$$
(B.18)

Using the FOC (B.15) and our guess, we get the consumption share:

$$\frac{C_t^*}{Y_t^*} = \frac{M_t^{\xi}}{\sum_{j=0}^{\infty} \alpha^j \beta^j M_{t+j}^{\xi}}$$
(B.19)

Hence, given the feasibility constraint, the investment share is given by (33).

Notice that this expression can be also found by replacing equation (B.18) into equation (B.16) which validates our guess. Finally, using (B.18) and (B.17), we obtain optimal carbon prices.

#### Corollary 6

*Proof.* For  $\xi = 0$ , or  $\xi > 0$  with  $M_t = M$ , from equation (35) it follows:

$$b_{t} = \frac{\sum_{j=0}^{\infty} \alpha^{j} \beta^{j+1}}{\sum_{j=0}^{\infty} \alpha^{j} \beta^{j}}$$
$$= \frac{\beta \sum_{j=0}^{\infty} \alpha^{j} \beta^{j}}{\sum_{j=0}^{\infty} \alpha^{j} \beta^{j}}$$
$$= \beta$$
(B.20)

Finally, using (34), we get then:

$$p_t^* = Y_t \sum_{i=1}^{\infty} \theta_i \beta^i$$

The last part of the corollary follows from the above equation.

#### Corollary 7

*Proof.* For i = 1 and  $\xi = 1$ , using the expression for the investment share (33) and the population growth factor  $M_{t+1}/M_t = m_t$ , we obtain:

$$K_{t+1} = Y_t \left[ \frac{m_t \sum_{j=1}^{\infty} \alpha^j \beta^j \prod_{i=2}^{j} m_{t+i}}{1 + m_t \sum_{j=1}^{\infty} \alpha^j \beta^j \prod_{i=2}^{j} m_{t+i}} \right]$$
(B.21)

Therefore,

$$\frac{dK_{t+1}}{dm_t} = Y_t \left[ \frac{\sum_{j=1}^{\infty} \alpha^j \beta^j \prod_{i=2}^{j} m_{t+i}}{\left( 1 + m_t \sum_{j=1}^{\infty} \alpha^j \beta^j \prod_{i=2}^{j} m_{t+i} \right)^2} \right] > 0$$
(B.22)

From (7) it follows that  $\frac{dY_{t+1}}{dK_{t+1}} > 0$ , then using the previous result, we can also claim that  $\frac{dY_{t+1}}{dm_t} > 0$ . Using (B.21) we know that  $\frac{dK_{t+i+1}}{dY_{t+i}} > 0$  and from (7) that  $\frac{dY_{t+i+1}}{dK_{t+i+1}} > 0$ . The corollary thus holds for all i > 1 as well.

#### Corollary 8

*Proof.* For i = 0, notice that from (34-35) we know that if  $\frac{db_t}{dm_t} > 0$ , then  $\frac{dp_t}{dm_t} > 0$ . Applying the same procedure as before in the proof of corollary 7, it turns out that (35) can be written as follows

$$b_t = \frac{m_t^{\xi} \sum_{j=0}^{\infty} \alpha^j \beta^{j+1} \prod_{i=1}^j m_{t+i}^{\xi}}{1 + m_t^{\xi} \sum_{j=1}^{\infty} \alpha^j \beta^j \prod_{i=2}^j m_{t+i}^{\xi}}$$
(B.23)

The derivative of the last expression with respect to  $m_t$  is given by:

$$\frac{db_t}{dm_t} = \frac{\xi m_t^{\xi-1} \sum_{j=0}^{\infty} \alpha^j \beta^{j+1} \prod_{i=1}^j m_{t+i}^{\xi}}{\left[1 + m_t^{\xi} \sum_{j=1}^{\infty} \alpha^j \beta^j \prod_{i=2}^j m_{t+i}^{\xi}\right]^2} > 0$$
(B.24)

This implies that the corollary holds for i = 0. Likewise, we can show that  $\frac{\partial p_t}{\partial m_{t+i}} > 0$  for all i > 0.

#### Corollary 9

*Proof.* Since the carbon tax-GDP ratio  $\frac{p_t}{Y_t}$  depends on the discount factor  $b_t$  and this is a function of  $m_t$ , we just need to show that if  $m_t > m_{t+1}$ , then  $b_t - b_{t+1} > 0$ . Define  $\tilde{x}_t$  as

$$\tilde{x}_t \equiv m_t^{\xi} \sum_{j=0}^{\infty} \alpha^j \beta^{j+1} \prod_{i=1}^j m_{t+i}^{\xi}$$
(B.25)

Therefore, using (B.23) we obtain

$$b_{t} - b_{t+1} = \frac{\tilde{x}_{t}}{1 + \alpha \tilde{x}_{t}} - \frac{\tilde{x}_{t+1}}{1 + \alpha \tilde{x}_{t+1}} = \frac{\tilde{x}_{t} - \tilde{x}_{t+1}}{(1 + \alpha \tilde{x}_{t})(1 + \alpha \tilde{x}_{t+1})}$$
(B.26)  
> 0

as long as  $\tilde{x}_t > \tilde{x}_{t+1} > 0$ . Since  $\tilde{x}$  is a strictly increasing function of  $m_t$ , if  $m_t > m_{t+1}$ , then  $b_t > b_{t+1}$ , and we get the result.

### C. Balanced growth in the RCK Model with CES utility

The objective function  $W_t$  is given by:

$$\max_{\{c_t\}} W_t = \sum_t \beta^t M_t^{\xi} \frac{(C_t/M_t)^{1-\eta}}{1-\eta}$$
(C.1)

where  $C_t$  denotes consumption,  $M_t$  represents total population in this economy,  $\beta$  is discount factor,  $\xi$  is an indicator parameter, and  $\eta$  is the elasticity of marginal utility (the inverse of the elasticity of intertemporal substitution). The social planner maximizes (C.1) subject to the same economic constraints as in the main text. Let  $\lambda_t$  be the Lagrange multiplier associated to the goods constraint, thus

$$\beta^t M_t^{\xi - 1 + \eta} C_t^{-\eta} = \lambda_t \tag{C.2}$$

In balanced growth, population growth  $m = \frac{M_{t+1}}{M_t}$  is constant, and consumption  $C_t$ , output  $Y_t$  and capital  $K_t$  grow at the same rate  $\gamma m$ , where  $\gamma$  is the per capita income growth, thus:

$$\gamma m = \frac{Y_{t+1}}{Y_t} = \frac{C_{t+1}}{C_t} = \frac{K_{t+1}}{K_t}$$
(C.3)

The discount factor (26) is defined through:

$$b_t \equiv \frac{\lambda_{t+1}}{\lambda_t} \frac{Y_{t+1}}{Y_t} \tag{C.4}$$

Plugging (C.2) into (C.4), we get:

$$b_{t} = \frac{\beta^{t+1} M_{t+1}^{\xi - 1 + \eta} C_{t+1}^{-\eta}}{\beta^{t} M_{t}^{\xi - 1 + \eta} C_{t}^{-\eta}} \frac{Y_{t+1}}{Y_{t}}$$

$$= \beta \left(\frac{M_{t+1}}{M_{t}}\right)^{\xi - 1 + \eta} \left(\frac{C_{t+1}}{C_{t}}\right)^{-\eta} \frac{Y_{t+1}}{Y_{t}}$$
(C.5)

which becomes in balanced growth:

$$b = \beta m^{\xi - 1 + \eta} (m\gamma)^{1 - \eta}$$
  
=  $\beta m^{\xi} \gamma^{1 - \eta}$  (C.6)

## **D.** Calculation of $g_t$ as in (18)

Consider the sequence  $\theta_i$  as in (38), and the policy function  $g_t$  as in (18):

$$g_t = \sum_{i=1}^{\infty} b_t^{t+i} \theta_i. \tag{D.1}$$

First, we decompose the response series  $\theta_i$  (38) into its parts

$$\theta_i = \sum_k \theta_{i,k}, \tag{D.2}$$

$$\theta_{i,k} = \sum_{j} a_j b_k \varepsilon_k \frac{(1-\eta_j)^i - (1-\varepsilon_k)^i}{\varepsilon_k - \eta_j}, \qquad (D.3)$$

and note that these parts satisfy the recursive equation

$$\theta_{i,k} = (1 - \varepsilon_k)\theta_{i-1,k} + \sum_j (1 - \eta_j)^{i-1}.$$
(D.4)

We now define the auxiliary variable

$$x_{j,t} \equiv \sum_{i=0}^{\infty} b_t^{t+i} (1-\eta_j)^i$$

$$= 1 + (1-\eta_j) b_{t+1} \sum_{i=1}^{\infty} b_{t+1}^{t+i} (1-\eta_j)^{i-1}$$

$$= 1 + (1-\eta_j) b_{t+1} x_{j,t+1},$$
(D.5)

which helps us to calculate  $g_{k,t}$  recursively ( $g_t = \sum_k g_{k,t}$ )

$$g_{k,t} \equiv \sum_{i=1}^{\infty} b_t^{t+i} \theta_{i,k}$$
(D.6)  
=  $(1 - \varepsilon_k) \sum_{i=1}^{\infty} b_t^{t+i} \theta_{i-1,k} + b_{t+1} \sum_j \sum_{i=1}^{\infty} b_{t+1}^{t+i} (1 - \eta_j)^{i-1}$   
=  $(1 - \varepsilon_k) b_{t+1} \sum_{i=1}^{\infty} b_{t+1}^{t+i} \theta_i + b_{t+1} \sum_j x_{j,t+1}$   
=  $(1 - \varepsilon_k) b_{t+1} g_{k,t+1} + b_{t+1} \sum_j x_{j,t+1}.$ 

Thus, if we assume that demography is in steady state at the final year, t = T, then  $b_t$ ,  $x_t$  and  $g_{k,t}$  are also stationary from that period onwards, and we can calculate the variables  $x_t$  and  $g_t$ , at that period, through

$$x_{j,t} = \frac{1}{1 - b_t (1 - \eta_j)}$$
(D.7)

$$g_{k,t} = \frac{b_t \sum_j x_{j,t}}{1 - b_t (1 - \varepsilon_k)}$$
(D.8)

The recursive equations are then used to calculate  $(x_t, g_t)$  backwards.

## E. Calculation of $b_t$ as defined in (36)

Notice that (36) can be written as follows:

$$b_t = \frac{\sum_{j=0}^{\infty} \alpha^j \beta^{j+1} (\gamma_t^{t+j+1})^{1-\eta} (m_t^{t+j+1})^{\xi}}{\sum_{j=0}^{\infty} \alpha^j \beta^j (\gamma_t^{t+j})^{1-\eta} (m_t^{t+j})^{\xi}}$$
(E.1)

Define  $z_t$  as:

$$z_{t} \equiv \sum_{i=0}^{\infty} \alpha^{j} \beta^{j} (\gamma_{t}^{t+j})^{1-\eta} (m_{t}^{t+j})^{\xi}$$
  
=  $1 + \alpha \beta \gamma_{t}^{1-\eta} m_{t}^{\xi} \sum_{j=0}^{\infty} \alpha^{j} \beta^{j} (\gamma_{t+1}^{t+j+1})^{1-\eta} (m_{t+1}^{t+j+1})^{\xi}$   
=  $1 + \alpha \beta \gamma_{t}^{1-\eta} m_{t}^{\xi} \sum_{j=1}^{\infty} \alpha^{j-1} \beta^{j-1} (\gamma_{t+1}^{t+j})^{1-\eta} (m_{t+1}^{t+j})^{\xi}$   
=  $1 + \alpha \beta \gamma_{t}^{1-\eta} m_{t}^{\xi} z_{t+1}$  (E.2)

Therefore,

$$b_t = \frac{\beta \gamma_t^{1-\eta} m_t^{\xi} z_{t+1}}{z_t} \tag{E.3}$$

As shown in Appendix C, in balanced growth the last expression collapses to  $b = \beta \gamma^{1-\eta} m^{\xi}$ . But now, we want to solve (E.3) recursively, and we use our auxiliary variable  $z_t$  to do so. Assume that population m and productivity  $\gamma$  growth rates are in steady state at t = T, our final year, so that  $b_T$  and  $z_T$  are stationary from that period onwards. Thus, we obtain

$$z_T = \frac{1}{1 - \alpha \beta \gamma_T^{1-\eta} m_T^{\xi}} \tag{E.4}$$

And we utilize the last equation to calculate  $(z_t, b_t)$  backwards.